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ABSTRAK

Sirup glukosa diperoleh melalui proses hidrolisis pati yang sering dimanfaat-
kan di industri makanan serta industri farmasi. Bahan dasar untuk sirup glukosa
meliputi: jagung, singkong, sagu, dan lain-lain. Bahan dasar untuk pembuatan sirup
glukosa sangat banyak tersedia di Indonesia. Produksi jagung di dalam negeri men-
capai rata-rata 22,5 juta ton setiap tahunnya. Di Indonesia terdapat tiga pabrik yang
menghasilkan produk tersebut dengan kapasitas mencapai 173. 000 ton per tahun,
serta melakukan impor sebanyak 102. 489 ton per tahun. Tujuan dari pra rancang
pabrik yang akan didirikan adalah untuk memenuhi kebutuhan sirup glukosa di da-
lam negeri, serta mengurangi ketergantungan pada impor dan menciptakan lapan-
gan kerja baru. Proses pembuatan sirup glukosa mencakup beberapa metode, yaitu:
hidrolisis pati menggunakan enzim, hidrolisis pati dengan asam, dan hidrolisis pati
menggunakan kombinasi asam dan enzim. Dalam pra rekayasa ini, kami memilih
untuk melakukan hidrolisis pati jagung menggunakan berbagai enzim. Langkah-
langkah dalam proses hidrolisis pati menggunakan enzim terbagi menjadi tiga fase,
yaitu: persiapan bahan, reaksi, dan pemurnian produk. Kapasitas yang diharapkan
adalah 100. 000 Ton per tahun di Kabupaten Tuban, Jawa Timur. Oleh karena itu,
pembangunan pabrik sirup glukosa akan mengurangi kebutuhan impor serta men-
ciptakan lapangan kerja baru.

Kata Kunci : Sirup Glukosa, Hidrolisis, Pati, Jagung
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ABSTRACT

Glucose syrup is obtained through starch hydrolysis process which is often
utilized in food industry as well as pharmaceutical industry. Basic materials for
glucose syrup include: corn, cassava, sago, and others. The basic ingredients for
glucose syrup are widely available in Indonesia. Maize production in the country
reaches an average of 22.5 million tons annually. In Indonesia, there are three
factories that produce this product with a capacity of 173. 000 tons per year, and
imports as much as 102. 489 tons per year. The purpose of the pre-design of the
plant to be established is to meet the needs of glucose syrup in the country, as well
as reduce dependence on imports and create new jobs. The process of making
glucose syrup includes several methods, viz: starch hydrolysis using enzymes,
starch hydrolysis with acids, and starch hydrolysis using a combination of acids
and enzymes. In this pre-engineering, we chose to perform corn starch hydrolysis
using various enzymes. The steps in the process of starch hydrolysis using enzymes
are divided into three phases, viz: material preparation, reaction, and product
purification. The expected capacity is 100,000 tons per year in Tuban Regency,
East Java. Therefore, the construction of a glucose syrup plant will reduce the need
for imports and create new jobs. Using corn starch as much as 8.732 kg per hour,
o-Amylase enzyme as much as 8.16 kg per hour, and glucoamylase as much as 8.12
kg per hour, to support the operation of the plant required water, electricity, fuel
and labor.Economic analysis BEP: 23.889%, POT: 1.69 years, SDP: §
6,850,121.19 so that this plant is very feasible to operate.
Keywords.: Glucose Syrup, Hydrolysis, Starch, Maize
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